\documentclass{article}

%%%%% do not modify below %%%%%
\usepackage{fullpage}
\usepackage{xspace}
\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{amsthm}

\usepackage{tikz}
\usetikzlibrary{positioning}
\usetikzlibrary{fit}
\usetikzlibrary{calc}
\usetikzlibrary{backgrounds}
\usetikzlibrary{shapes}
\usetikzlibrary{patterns}
\usetikzlibrary{matrix}
\usetikzlibrary{intersections}
\usetikzlibrary{decorations}
\usetikzlibrary{arrows.meta}
%%%%% do not modify above %%%%%




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%% Students - Fill out below %%%

\newcommand{\studentName}{Your name here}

%%% Student Information - Done %%%

%%% You may add additional packages, macros, etc., in this section.







%%% End local packages and macros
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%





%%%%% do not modify below %%%%%
\usepackage{hyperref}
\hypersetup{pdfstartview=FitH,colorlinks,linkcolor=blue,filecolor=blue,citecolor=blue,urlcolor=blue,breaklinks}

\usepackage[nameinlink,capitalize]{cleveref}




\title{In-class Report Template}
\date{\today}
\author{\studentName}

\usepackage[style=trad-alpha]{biblatex}
\addbibresource{local.bib} % default bib style
\renewcommand{\multicitedelim}{\addcomma\space}
%%%%% do not modify above %%%%%



\begin{document}
\maketitle

\section{Problem(s) given in class}
\emph{Brief list of problem(s) given during class you were asked to solve/think about.}

\section{How you or your group approached solving these problems}
\emph{What did you or your group do to solve these problems? Did you look things up online? Did you remember from another course? Did you write some code, write on paper, or use the board during your thought process and discussion?}

\section{Challenges you faced solving these problems}
\emph{What was difficult for you when trying to solve these problems? Did you have trouble analyzing the run time? Was the algorithm to solve them difficult? Did you not know where to start solving the problem?}

\section{Insights gained/things you learned trying to solve these problems}
\emph{What did you learn while trying to solve the problem? Was it something new, or something you learned before and recalled? Did a group member, or another person/group in-class help you find the solution?}

\section{Final solution to problems}
\emph{This does not need to be in-depth, it can be a rapid-fire set of answers, or pseudo-code if we did in-class coding. For example, being asked to discuss a sorting algorithm, including high-level overview of the algorithm, along with best/worst/average case runtimes.}

\section{List of group members}
\emph{A list of names of your other group members.}

%%%%% Bibliography %%%%%
%%% Please put all bibliography entries into local.bib, which is blank by default

%%%% Uncomment the last two lines if you are citing things in the bibliography.
%\newpage
%\printbibliography


\end{document}

